Author’s response to reviews

Title: BRCA Mutations in a Cohort of Iraqi Patients Presenting to a Tertiary Referral Center

Authors:

Farra Chantal (cfarra@hotmail.com)

Dagher Christelle (cd19@aub.edu.lb)

Hamadeh Lama (lh64@aub.edu.lb)

Nagi El Saghir (ns23@aub.edu.lb)

Deborah Mukherji (dm25@aub.edu.lb)

Version: 2 Date: 22 Jul 2019

Author’s response to reviews:

Thank you for your review and opportunity to submit a revised manuscript.

We have reviewed the data present in the genetics charts - since this was a retrospective review of patients who are living in Iraq we do not have complete data. As far as we can ascertain, all patients had a diagnosis of breast cancer, some had a family history of breast or ovarian cancer in addition (included in the results section and table 1).

All available data have been included.

Kind regards

Dr Deborah Mukherji